\documentclass{article}
\usepackage{graphicx}
\usepackage[landscape]{geometry}
\usepackage{geometry}
\geometry{a4paper,scale=0.9}


\begin{document}

\begin{table*}
\begin{center}
\caption[]{Basic parameters for 13 previously known pulsars, including right ascension, declination, barycentric periods, period derivatives, dispersion measures, with an $1\sigma$ uncertainty given in brackets, together with the epoch for the period, data span and weighted RMS residuals. For each pulsar, previously published parameters in references are given in the first line, and our new measurements are given in the second line for every pulsar. To compare with previously published parameters, our timing solutions are presented in the TDB unit.
}
\label{knowntab1}
 \tabcolsep 4pt 
 \footnotesize
 \begin{tabular}{llllllrlcl}
  \hline
  %\noalign{\smallskip}
Name          & Ref.      & RA                & DEC               & $P$                  & $\dot{P}$    & DM                & Epoch & Span  & $W_{rms}$ \\
              &           & (hh:mm:ss)        & (dd:mm:ss)        & (s)                & ($10^{-15}$ s/s)     & ($\rm cm^{-3}$pc) & (MJD) & (yr)  & (ms)  \\
\hline\noalign{\smallskip}
J1852$-$0000  & [1]        &  18:52:40.167(9) &  $-$00:00:25.5(3) &  1.92066632921(2) &  251.9666(3)  & 593(1)  &    58205.0 &5.7\\    
              & this work  &  18:52:40.15(1)  &  $-$00:00:26.3(7)  &  1.92068580196(3)  &  251.933(1)  &  593.1(3)  &  59403.0  &  3.74  &  0.523 \\
J1852$+$0008$^*$  & [2]        & 18:52:42.78(3)    & $+$00:08:09.6(8)  & 0.467894113075(20) & 5.679(7)   &  254.9(18)  & 52584.0    & 1.50  &    \\
              & this work  &  18:52:42.802(2)  &  $+$00:08:09.6(1)  &  0.4678973753676(9)  &  5.68435(6)  &  255.40(6)  &  59228.0  &  2.78  &  0.174 \\
J1853$+$0029$^*$  & [1]        & 18:53:17.745(17)  & $+$00:29:23.8(7)  &  1.8767576226(3)  &  2.431(2)   &  232(4)     &57826.0&3.5 & \\
              & this work  & 18:53:17.783(4)  &  $+$00:29:22.1(4)  &  1.87675787185(1)  &  2.428(2)  &  227.39(8)  &  59013.0  &  1.60  &  0.208 \\
J1858$+$0241$^*$  & [2]        & 18:58:53.81(14) &   $+$02:41:38(6)   &  4.6932329333(12)   &  24.32(9)  &  336(15)    &  52111.0 & 2.31   \\
              & this work  &  18:58:53.81(2)  &  $+$02:41:37.5(5)  &  4.69324828044(5)  &  24.274(6)  &  325.9(5)  &  59425.0  &  2.88  &  0.794 \\
J1901$+$0320$^*$  & [2]        &  19:01:03.01(9)  & $+$03:20:18(4)     & 0.63658447822(8) &    0.52(3)   &  393(7) & 52503.0 &1.22 & \\
              & this work  &  19:01:03.082(6)  &  $+$03:20:17.5(2)  &  0.63658476124(1)  &  0.5096(4)  &  394.8(3)  &  59428.0  &  2.87  &  0.296 \\
J1901$+$0510$^*$  & [2]        &  19:01:57.85(11) & $+$05:10:34(4)     &  0.61475669408(12) &  31.10(4)  &  429(7)    &   52618.0 & 1.15 \\
              & this work  &  19:01:57.90(5)  &  $+$05:10:36.5(7)  &  0.61477450398(2)  &  31.113(2)  &  435.5(2)  &  59243.0  &  1.91  &  0.618 \\
J1905$+$0600$^*$  & [2]        & 19:05:04.35(5)    & $+$06:00:59.9(14) & 0.441209731966(18) & 1.1123(10) &  730.1(19)  & 52048.0   & 3.17  &    \\
              & this work  &  19:05:04.3352(8)  &  $+$06:01:00.65(2)  &  0.4412104290545(3)  &  1.10719(2)  &  728.44(3)  &  59395.0  &  2.72  &  0.074 \\
J1905$+$0902$^*$  & [3],[4]    & 19:05:19.535(2)   & $+$09:02:32.49(8) & 0.2182529126846(9) & 3.49853(8) &  433.4(1)   & 54570.0  &   &    \\
              & this work  &  19:05:19.5414(8)  &  $+$09:02:32.39(2)  &  0.2182543240059(3)  &  3.499036(9)  &  433.497(7)  &  59355.0  &  3.35  &  0.042 \\
J1906$+$0649$^*$  & [2]        & 19:06:11.97(3)    & $+$06:49:48.1(10) & 1.28656437956(10)  & 0.152(5)   &  249(4)     & 52317.0   & 2.80  &    \\
              & this work  &  19:06:12.003(2)  &  $+$06:49:48.5(2)  &  1.286564468927(6)  &  0.1498(4)  &  250.16(9)  &  59200.0  &  2.47  &  0.292 \\
J1927$+$1852$^*$  & [5],[6],[7]& 19:27:10.422(8)   & $+$18:52:08.5(2)  & 0.482766273821(7)  & 0.116(1)   &  264.5(4)   & 51600.0   &   &    \\
              & this work  &  19:27:10.414(5)  &  $+$18:52:08.62(9)  &  0.482766349761(2)  &  0.11606(6)  &  264.66(6)  &  59310.0  &  4.05  &  0.190 \\
J1930$+$1408  & [1],[8]    & 19:30:18.9526(18) & $+$14:08:55.39(5) & 0.425720327378(5)  & 0.00190(1) &  210.87(13) & 56885.0   & 8.6      &    \\  
              & this work  & 19:30:18.949(5)  &  $+$14:08:55.3(1)  &  0.425720327769(1)  &  0.00194(9)  &  212.11(9)  &  59388.0  &  3.17  &  0.227 \\
J1936$+$2042  & [1],[8]    & 19:36:27.42(2)    & $+$20:42:04.5(4)  & 1.39072342303(15)  & 49.3744(14)&  197.4(5)   &  56065.0  & 5.8  &    \\
              & this work  &  19:36:28.702(1)  &  $+$20:41:26.62(3)  &  1.390726665137(2)  &  5.2926(1)  &  195.11(3)  &  59441.0  &  2.96  &  0.080 \\
J1954$+$2923  & [9],[10]   & 19:54:22.554(2)  &  $+$29:23:17.29(4)  &  0.4266767865302(4)  & 0.001711(3) & 7.932(7)  & 48719.0  & 21.4 &      \\
              & this work  &  19:54:22.5107(4)  &  $+$29:23:16.15(1)  &  0.4266767880541(2)  &  0.001663(6)  &  7.932(7)  &  59336.0  &  4.25  &  0.040 \\
\noalign{\smallskip}\hline\noalign{\smallskip}
\end{tabular}
\end{center}
{$^*$ Pulsars with improved parameters.}
References: 
[1]: \citet{2022ApJ...924..135P}; [2]: \citet{2004MNRAS.352.1439H}; [3]: \citet{2006ApJ...637..446C}; [4]: \citet{2013ApJ...772...50N}; [5]: \citet{1975ApJ...201L..55H}; [6]: \citet{2002AJ....123.1750L}; [7]: \citet{2021RAA....21..107H}; [8]: \citet{2015ApJ...812...81L}; [9]: \citet{1970Natur.227.1123D}; [10]: \citet{2004MNRAS.353.1311H}.
\end{table*}


\begin{table}
\begin{center}
\caption[]{Best-fit parameters of timing models of 30 isolated pulsars newly discovered in the GPPS survey and derived parameters. Columns are set for pulsar name,   previous temporary name in \citet{2021RAA....21..107H} together with the GPPS discovery number, the flux density at 1.25 GHz, right ascension (2000), declination (2000), the  barycentric periods, period derivatives, dispersion measures, the epoch for the period, the data span, weighted RMS residuals, the derived parameters for the characteristic age, surface magnetic field strength, and spin-down luminosity. The 1$\sigma$ uncertainty is given in brackets, and the results are presented in TCB unit.}  
\label{GPPStab1}
 \tabcolsep 3pt  
 \footnotesize
 \begin{tabular}{llcllllrcccccc}
   \hline\noalign{\smallskip}
PSR Name & Previous name: gpps No.  &  $S_{\rm 1.25GHz}$&  RA(2000)         & DEC(2000)         & $P$                    &$\dot{P}$& DM               & Epoch    & Span  & $W_{rms}$ &log[$\rm \tau$/yr]       & log[B/G]            &  log[$\rm \dot{E}$/erg $\rm s^{-1}$]\\
              &                     &  ($\mu$Jy)        & (hh:mm:ss)       & (dd:mm:ss)        & (s)                  &($10^{-15}$ s $\rm s^{-1}$) & ($\rm cm^{-3}$pc)& (MJD)    & (yr)  & (ms) \\
  \hline\noalign{\smallskip}
J1837$+$0033  &J1837$+$0033g: gpps0051 & 33.1  &  18:37:41.100(4)  &  $+$00:33:46.1(1)  &  0.418131244964(1)  &  0.02273(6)  &  187.4(1)  &  59477.0  &  3.26  &  0.248  & 8.46  &  10.99  &  31.09 \\
J1849$+$0002  &J1849$+$0001g: gpps0093 & 16.3  &  18:49:22.557(5)  &  $+$00:02:17.0(2)  &  0.525650459766(2)  &  13.1274(1)  &  188.2(3)  &  59511.0  &  3.33  &  0.444  & 5.80  &  12.42  &  33.55 \\
J1849$+$0010  &J1849$+$0009g: gpps0096 & 17.8  &  18:49:42.670(6)  &  $+$00:10:00.4(1)  &  1.318430596396(5)  &  87.8234(3)  &  505.1(2)  &  59524.0  &  3.26  &  0.365  & 5.38  &  13.04  &  33.18 \\
J1849$-$0013  &J1849$-$0014g: gpps0027 & 149.8 &  18:49:26.996(3)  &  $-$00:13:46.8(1)  &  0.491754129798(2)  &  17.2224(1)  &  345.9(1)  &  59331.0  &  2.34  &  0.243  & 5.66  &  12.47  &  33.76 \\
J1850$-$0002  &J1850$-$0002g: gpps0097 & 60.7  &  18:50:02.340(9)  &  $-$00:02:09.7(3)  &  0.893355410120(6)  &  6.4624(7)  &  541.7(4)  &  59396.0  &  2.70  &  0.636  & 6.34  &  12.39  &  32.55 \\
J1850$-$0020  &J1850$-$0020g: gpps0018 & 66.3  &  18:50:01.658(5)  &  $-$00:20:37.2(2)  &  1.574737324031(7)  &  24.0214(8)  &  602.8(2)  &  59331.0  &  2.34  &  0.380  & 6.02  &  12.79  &  32.39 \\
J1852$+$0018  &J1852$+$0018g: gpps0103 & 45.1  &  18:52:26.742(3)  &  $+$00:18:42.4(1)  &  0.318762753335(1)  &  27.09983(8)  &  454.8(1)  &  59228.0  &  2.78  &  0.218  & 5.27  &  12.47  &  34.52 \\
J1852$-$0024  &J1852$-$0024g: gpps0042 & 58.2  &  18:52:14.1296(8)  &  $-$00:24:17.67(3)  &  0.3554290457076(3)  &  0.16460(2)  &  292.31(8)  &  59438.0  &  2.82  &  0.049  & 7.53  &  11.39  &  32.16 \\
J1852$-$0033  &J1852$-$0033g: gpps0173 & 22.7  &  18:52:26.526(3)  &  $-$00:33:39.71(9)  &  1.369012367933(4)  &  7.9154(3)  &  321.6(1)  &  59676.0  &  2.30  &  0.216  & 6.44  &  12.52  &  32.09 \\
J1852$-$0039  &J1852$-$0040g: gpps0177 & 79.7  &  18:52:13.195(5)  &  $-$00:39:44.8(2)  &  0.802905417851(4)  &  0.3783(3)  &  354.0(3)  &  59676.0  &  2.30  &  0.368  & 7.53  &  11.75  &  31.46 \\
J1853$+$0014  &J1853$+$0013g: gpps0030 & 35.2  &  18:53:52.502(3)  &  $+$00:14:12.8(5)  &  0.928595784010(3)  &  1.1422(3)  &  309.6(2)  &  59490.0  &  3.18  &  0.217  & 7.11  &  12.02  &  31.75 \\
J1853$+$0021  &J1853$+$0023g: gpps0089 & 19.4  &  18:53:07.748(5)  &  $+$00:21:31.2(4)  &  0.576939240923(6)  &  0.0650(4)  &  205.6(3)  &  59228.0  &  2.78  &  0.378  & 8.15  &  11.29  &  31.13 \\
J1855$+$0234  &J1855$+$0235g: gpps0148 & 14.9  &  18:55:52.374(2)  &  $+$02:34:55.58(6)  &  0.983029834436(3)  &  0.0215(4)  &  103.71(6)  &  59447.0  &  1.59  &  0.109  & 8.86  &  11.17  &  29.95 \\
J1856$+$0211  &J1856$+$0211g: gpps0158 & 32.4  &  18:56:59.42(1)  &  $+$02:11:10.9(4)  &  9.89009157558(8)  &  0.78(1)  &  124.5(5)  &  59607.0  &  2.64  &  0.719  & 8.30  &  12.45  &  28.50 \\
J1857$+$0214  &J1857$+$0214g: gpps0010 & 59.6  &  18:57:07.406(8)  &  $+$02:14:37.2(3)  &  0.333917722223(2)  &  1.6059(2)  &  982.0(7)  &  59427.0  &  3.63  &  0.899  & 6.52  &  11.87  &  33.23 \\
J1857$+$0224  &J1857$+$0224g: gpps0072 & 35.2  &  18:57:10.706(5)  &  $+$02:24:43.2(2)  &  0.875874011838(6)  &  0.0615(5)  &  398.0(5)  &  59410.0  &  2.74  &  0.319  & 8.35  &  11.37  &  30.56 \\
J1859$+$0430  &J1859$+$0430g: gpps0015 & 18.5  &  18:59:08.750(5)  &  $+$04:30:18.7(1)  &  0.336324930220(2)  &  0.053(1)  &  782.9(4)  &  59104.0  &  1.14  &  0.154  & 8.00  &  11.13  &  31.74 \\
J1903$+$0845  &J1903$+$0845g: gpps0048 & 30.4  &  19:03:39.6082(9)  &  $+$08:45:58.00(2)  &  0.1531502678628(1)  &  0.091961(7)  &  129.56(6)  &  59503.0  &  3.10  &  0.085  & 7.42  &  11.08  &  33.00 \\
J1903$+$0851  &J1903$+$0851g: gpps0101 & 52.9  &  19:03:04.53(2)  &  $+$08:51:28.4(1)  &  1.23188198312(1)  &  27.1523(5)  &  83.0(1)  &  59567.0  &  2.75  &  0.554  & 5.86  &  12.77  &  32.76 \\
J1904$+$0853  &J1904$+$0852g: gpps0009 & 56.8  &  19:04:55.29234(9)  &  $+$08:53:09.823(2)  &  0.0061974296377438(4)  &  0.00001009(3)  &  195.159(3)  &  59355.0  &  3.35  &  0.007  & 9.99  &   8.40  &  33.22 \\
J1905$+$0656  &J1905$+$0656g: gpps0008 & 60.7  &  19:05:44.343(6)  &  $+$06:56:35.0(2)  &  2.51176254691(2)  &  0.387(2)  &  24.2(2)  &  59267.0  &  2.10  &  0.381  & 8.01  &  12.00  &  29.98 \\
J1905$+$0758  &J1905$+$0758g: gpps0043 & 35.2  &  19:05:50.964(3)  &  $+$07:58:52.62(5)  &  1.192648616445(2)  &  0.0459(2)  &  197.1(1)  &  59429.0  &  2.83  &  0.149  & 8.61  &  11.37  &  30.03 \\
J1905$+$0935  &J1905$+$0936g: gpps0153 & 60.8  &  19:05:26.088(4)  &  $+$09:35:59.41(9)  &  1.634497451794(4)  &  1.5008(4)  &  417.96(4)  &  59650.0  &  2.30  &  0.141  & 7.24  &  12.20  &  31.13 \\
J1906$+$0757  &J1906$+$0757g: gpps0102 & 28.9  &  19:06:41.7449(3)  &  $+$07:57:01.842(4)  &  0.05718905678761(1)  &  0.000116(1)  &  79.104(6)  &  59430.0  &  2.83  &  0.011  & 9.89  &   9.42  &  31.39 \\
J1917$+$0748  &J1916$+$0748g: gpps0175 & 11.9  &  19:17:00.899(8)  &  $+$07:48:43.9(2)  &  0.867880485703(5)  &  0.1981(6)  &  153.6(2)  &  59607.0  &  2.54  &  0.353  & 7.84  &  11.62  &  31.08 \\
J1917$+$0743  &J1917$+$0743g: gpps0167 & 32.8  &  19:17:25.912(2)  &  $+$07:43:16.69(4)  &  0.8134532288005(8)  &  0.04485(8)  &  198.39(4)  &  59652.0  &  2.29  &  0.072  & 8.46  &  11.29  &  30.52 \\
J1926$+$1631  &J1926$+$1631g: gpps0053 & 78.6  &  19:26:23.9083(6)  &  $+$16:31:27.78(1)  &  0.6783666636602(3)  &  0.03267(2)  &  196.73(3)  &  59487.0  &  3.21  &  0.050  & 8.52  &  11.18  &  30.62 \\
J1928$+$1853  &J1928$+$1852g: gpps0143 & 27.9  &  19:28:03.448(6)  &  $+$18:53:04.9(1)  &  0.792806958052(3)  &  0.0743(2)  &  290.0(3)  &  59631.0  &  2.40  &  0.250  & 8.23  &  11.39  &  30.77 \\
J1930$+$1357  &J1930$+$1357g: gpps0050 & 16.0  &  19:30:11.991(2)  &  $+$13:57:06.43(3)  &  0.3235617566079(4)  &  0.02723(3)  &  184.8(1)  &  59420.0  &  3.35  &  0.110  & 8.28  &  10.98  &  31.50 \\
J2017$+$2819  &J2017$+$2819g: gpps0044 & 61.0  &  20:17:23.297(2)  &  $+$28:19:14.26(3)  &  1.832558917299(6)  &  0.3580(4)  &  66.0(2)  &  58997.0  &  2.34  &  0.074  & 7.91  &  11.91  &  30.36 \\
  \noalign{\smallskip}\hline
\end{tabular}
\end{center}
\end{table}

\end{document}
